Author's response to reviews

Title: Red ginseng abrogates oxidative stress via mitochondria protection mediated by LKB1-AMPK pathway

Authors:

Guang-Zhi Dong (dgz77@hanmail.net)
Eun Jeong Jang (lemon3321@hanmail.net)
Seung Ho Kang (ksh@sunlin.ac.kr)
Il Je Cho (skek023@dhu.ac.kr)
Sang Chan Kim (sckim@dhu.ac.kr)
Young Woo Kim (ywkim@dhu.ac.kr)

Version: 3 Date: 20 February 2013

Author's response to reviews: see over
RE: MS # 1960475704805015
Title: Red ginseng abrogates oxidative stress via mitochondria protection mediated by LKB1-AMPK pathway

Dear Editor:

Enclosed please find the revised manuscript by Dong et al entitled as above. We revised the manuscript according to the referee’s helpful comments. Specific details are described in the letter for reviewer’s comments.

I would appreciate your consideration.

Sincerely Yours,

Young Woo Kim, Ph.D.
Medical Research Center for Globalization of Herbal Formulation
College of Oriental Medicine
Daegu Haany University
Daegu 706-828, Korea
Tel: +82-53-770-2341; E-mail: ywkim@dhu.ac.kr
The authors express sincere thanks to the referee for the valuable comments. We revised the manuscript according to the referee’s helpful comments.

**EDITORIAL REQUIREMENT**
While we note that a number of corrections have been made in the quality of English in your manuscript, we still feel that it would benefit from some improvements being made to the style and grammar.

**Answer:** As editors recommended, we improved the quality of written English by Editing Program.